 
CENWP-OP-B                                                                                               15 November 1999

MEMORANDUM FOR CENWP-OP, ATTN:  Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 45 of 1999, which covers the period 7 through 13 November.
1.  OPERATION SUMMARY.

1.1. Daily average river flow ranged from 130.2 to 140.7 kcfs.  Daily average tailwater elevation for the project ranged from 12.3’ to 13.5’ msl.  River temperature was 54° F all week.  Spill bays 1 and 18, which supply attraction flow for the Cascades Island and B-branch fish ladders respectively, remain open 4”. 

1.2.  First powerhouse Unit 4 remains out of service for rehab.  Units 5 and 6 are being operated according to index testing requirements.  Unit 10 remains out of service for stator rewind.  Unit 11 was taken out of service on 8 November for four year overhaul.  Unit 16 remains out of service for stator re-wedge.  The unit is scheduled to return to service on 18 November.

1.3. The first powerhouse ice and trash sluiceway remained open for debris removal this reporting week.  The second powerhouse ice and trash sluiceway gate was closed.

1.4. Second powerhouse fish units remained in service this reporting week except when placed on standby from approximately 0000-0300 to float trash if drawdown exceeds 1.0’.

2.  ACTIVITIES. 
2.1. STS Inspections. Five units’ STS’s were inspected this reporting week.  All screens were in good condition unless otherwise noted below.  See Table 1.

Unit
Date
Run Hours
Comments

11
11/8
940
Repaired two small rips in B-slot screen.

18
11/9
782


Table 1.  STS Inspection Data.


2.2. VBS Inspections. There were no VBS inspections this reporting week.

2.3. AWS Closures.  FV1-1 was closed 1500 - 1516 on 10 November for trash raking.  

2.4. Fish Salvage. Nine sturgeon and 2 lamprey were removed from Unit 11 draft tube on 8 November in preparation for the unit’s four-year overhaul.  All fish were released in good condition.

3.  RESEARCH. 

3.1. CRITFC/ODFW/WDFW AFC&MF Sampling.  Completed.

3.2. CRITFC Dam Angling. Completed.

3.3. NMFS/FFU PIT Tag and Video Monitoring. Completed.

3.4. NMFS/University of Idaho Lamprey Test Flume. Completed.

3.5. NMFS Lamprey Trapping and Radio Tracking. Completed.

3.6. PSMFC Smolt Monitoring. Completed.

3.7. WDFW Fish Counting.  WDFW (John Loch) using video counting at Bradford Island and Washington shore.  

4.  FISHWAYS.  Project biologists inspected on 8 and 9 November.  

4.1. Fish Passage Plan Observations. 

DATE
LOCATion
FPP VIOLATION
CAUSE
RESPONSE

All Week

Project biologists did not complete 3 inspections this week as required per the FPP.
Short-staffed and additional activities.
Operators continue to inspect fishways daily.

All Week
A-branch diffuser FG3-4.
Diffuser should be open at 14.4’ tailwater elevation and is stuck closed due to mechanical failure.
Diffuser gate is separated from its shaft and will not be repaired until the winter maintenance season. 
Other diffusers are being operated in place of FG3-4 depending on tailwater elevations.

All Week
A-branch diffuser FG3-6.
Diffuser should be open at 20.4’ tailwater elevation and is stuck closed due to mechanical failure. 
Diffuser gate is separated from its shaft and will not be repaired until the winter maintenance season.
Other diffusers are being operated in place of FG3-4 depending on tailwater elevations.

All Week
PH1 chain gate 10C.
Gate was not on sill as required per the FPP.
Gate was raised for unknown reason. 
No action taken.

All Week
OG 21
OG 21 should be closed.
Panel is missing, allowing water to pass freely when gate is closed.
Will be repaired this winter maintenance season.

All week
PH1 collection channel electronic velocity meter.
Meter was not working and should register 1.5-4.0 fps.
Unknown.  Meter is working intermittently.
No action taken.

4.2. Sensor Calibration.  Water level sensors were checked for calibration on 9 November.   

4.3. Avian Predation Status.  There are currently three avian lines down at PH1. 

4.4. Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.

5.  CONSTRUCTION UPDATE.
5.1. Second Powerhouse DSM channel and flume continue to operate in automatic control.  Orifice 11AS remains closed because of poor flow characteristics.  Add-in water supply remains closed.  The outfall remains in low outfall bypass.  DSM screen cleaners are controlled automatically and cycle every three hours.

                                                                                                     DEBORAH L. CHENOWETH

                                 

                                                              Operations Project Manager
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